Author’s response to reviews

Title: The differences in the consumption of proteins, fats and carbohydrates in the diet of pregnant women diagnosed with arterial hypertension or arterial hypertension and hypothyroidism

Authors:

Wioletta Waksmańska (wwaksmanska@ath.bielsko.pl)

Rafał Bobiński (rbobinski@ath.bielsko.pl)

Iabela Ulman-Włodarz (i.ulmanwlodarz@gmail.com)

Anna Pielesz (apielesz@ath.bielsko.pl)

Version: 3 Date: 16 Sep 2019

Author’s response to reviews:

Editor

Tovah Honor Aronin

Dear Ms Tovah Honor Aronin,

Thank you for your comments and critical response. All changes have been added to the manuscript according to the reviewers’ guidelines. All changes to the manuscript are indicated in the text by using green colour.

Please find below the response to the reviewers’ comments.

1. The title of the article has been corrected.

2. An additional conclusion was reached regarding the frequency of delivery according to Caesarean section in the group II and III.
3. The control group description was corrected as a group of healthy pregnant women with pregnancy without any complications and delivery on the due date.

A linguistic correction of the article has been made.

Yours sincerely

Wioletta Waksmańska